Check for updates

OPEN ACCESS

EDITED BY Bisma Mannan, University of Plymouth, United Kingdom

REVIEWED BY Abdul Naser Ibrahim Nour, An-Najah National University, Palestine Kun-Shan Wu, Tamkang University, Taiwan Simon Grima, University of Malta, Malta Shayut Pavapanunkul, Muhammadiyah University of Makassar, Indonesia Baogui Xin, Shandong University of Science and Technology. China

*CORRESPONDENCE Cancan Zhang ⊠ zhangcancan@dgcu.edu.cn

RECEIVED 14 March 2023 ACCEPTED 12 September 2023 PUBLISHED 05 October 2023

CITATION

Xie G, Zhang C, Fang Q, Tang X and Zhang Y (2023) Innovation, advertising, personal selling, and sustainability in the industries massively benefited from major public health emergencies: paradoxical evidence from China. *Front. Public Health* 11:1186026. doi: 10.3389/fpubh.2023.1186026

COPYRIGHT

© 2023 Xie, Zhang, Fang, Tang and Zhang. This is an open-access article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.

Innovation, advertising, personal selling, and sustainability in the industries massively benefited from major public health emergencies: paradoxical evidence from China

Guangying Xie¹, Cancan Zhang²*, Qianqian Fang³, Xiaole Tang³ and Yani Zhang⁴

¹School of Economics and Management, China University of Mining and Technology, Xuzhou, China, ²School of Business, Dongguan City University, Dongguan, China, ³School of Finance, Zhengzhou College of Finance and Economics, Zhengzhou, China, ⁴Chaobo Medical Materials Farming Cooperatives of Yangcheng, Jincheng, China

Background: While major public health emergencies have severe socioeconomic impacts, they may also present many opportunities for certain industries. For these industries that have benefited significantly (e.g., China' s healthcare industry), the traditional emphasis on improving business performance through increased investment in innovation, marketing and sustainability may face contextual applicability challenges.

Methods: We used the data of healthcare industry in China during Covid-19 and the methods of hierarchical regression, moderating effect test to analyze the impact of innovation, advertising, personal selling, and sustainability on healthcare firms' profitability. Three kinds of robust test including increasing the measurement range of variables, changing the data source and parameter estimation method, and Quantile regression are used.

Results: This paper finds that innovation, advertising, and environmental sustainability have significant negative impacts on profitability, while personal selling, social sustainability, and governance sustainability have significant positive impacts on profitability in the industries massively benefited from major public health emergencies. Besides, social sustainability can significantly moderate the relationship between innovation and profitability.

Conclusion: On one hand, for companies in industries that have benefited greatly from major public health emergencies, a shift in resource allocation from innovation, advertising, and environmental sustainability to personal selling, social sustainability, and governance sustainability may be more conducive to improving their profitability. On the other hand, for public health regulatory authorities, it is necessary to strengthen the supervision of sales representatives of health care enterprises, hospitals, public health organizations, etc., and appropriately subsidize the innovation of enterprises to enhance their innovation motivation.

KEYWORDS

innovation, environmental sustainability, social sustainability, governance sustainability, advertising, personal selling, profitability, COVID-19 pandemic

1. Introduction

The new crown pneumonia epidemic has lasted for more than 3 years, and has had a serious impact on the social economy, business operations and residents' lives. Academically, most of the existing literature is focused on how to survive from the dilemma, while neglects the research on the small number of industries and organizations that have gained significant market opportunities during this crisis. For the industries that have profited greatly from this major public health emergency, they are not only facing a massive increase in market demand, but also being in a scenario where most sectors of society are in recession or distress. This is a special situation in which "flowers wither away, and only this side of the scenery is good" in the process of economic crisis, which is significantly different from the rapid increase in market demand during the economic boom.

Resource allocation and optimization is a classic research topic in enterprise operation management and strategic decision-making (1-3). With limited resources, companies always need to reallocate resources according to the market environment to achieve optimal performance (4, 5). Innovation and marketing (including advertising and personal selling) are two traditional aspects that enterprises need to invest resources in. In general, both innovation and marketing investments can contribute to the profitability of a business. However, in an extremely favorable and extremely unfavorable market environment, the innovation and marketing behavior of enterprises may be largely affected as there may be more opportunism and survival pressure. If a company does not respond correctly to the changes in the external environment, it is likely to adversely affect its profitability. When the economy, society and most industries are suffering from major public health emergencies, it is interesting to study how to make decisions about the investment of innovation and marketing resources for industries that have benefited greatly. In particular, if these companies continue to invest more in innovation and marketing, will they be able to get more return on investment in this scenario? This is the main research gap that this paper fills in, which is seldomly studied in the past. On the other hand, the public has increasingly attached more importance to sustainability in recent years (6). Moreover, studies have shown that sustainability performs better in alleviating a company's financial crisis in times of economic crisis than the regular situations (7). Therefore, it is necessary and meaningful to take sustainability into consideration when analyzing the effects of resource allocation on profitability, which may complement the research on sustainability strategies in the context of major public health emergencies from the perspective of beneficiary industries and companies.

Healthcare companies, as providers of healthcare products and services, are also important members of the public health system. Benefiting from the industry opportunities brought by the new crown pneumonia epidemic, China's healthcare companies have achieved significant growth in net profit, return on assets, stock price and other financial performance indicators. Studying their innovation and marketing resource allocation and sustainable development behavior in this context can not only enrich the theory of organizational management and resource allocation of healthcare enterprises, but also have positive enlightenment for industry supervision and macrocontrol in the public health sector.

By using the data of China's healthcare industry, this paper mainly has the following findings and contributions. First, this study finds that for the industries massively benefited from major public health emergencies, innovation investment will have a negative impact on the profitability of current period. However, conventional wisdom is that innovation can improve many corporate financial performance indicators, including profitability (8, 9). This deviation, which is contrary to the traditional conclusion, provides a new contextual perspective for dialectical innovation investment decisions. Moreover, the paper furtherly finds that this adverse impact is negatively moderated by social sustainability. Second, for the two marketing tools including advertising and personal selling, this research finds that in the context of major public health emergency, advertising has a negative effect on profitability in the benefited industries. This also contradicts the conclusion that advertising investment can promote corporate performance proposed by most previous studies in general situations (10, 11), and provides a new contextual factor for a scientific view of the impact of advertising on corporate performance. Third, this paper confirms that the positive impact of personal selling, social sustainability, and governance sustainability, as well as the negative impact of environmental sustainability on corporate profitability remains applicable in industries that benefit greatly from major public health emergencies. We believe these findings can make a positive contribution to filling research gaps in innovation, marketing, and sustainability decision-making in industries that benefit from major public health emergencies, and to enriching the specific context of innovation, marketing, and sustainability research more broadly.

The rest of this paper is organized as follows: Section 2 presents a systematic literature review against the impact of innovation, advertising, personal selling, and sustainability on profitability and proposes corresponding hypotheses. Section 3 illustrates the data collection process, variables measurement, and research methodology. Section 4 shows the statistical results and Section 5 discusses the findings. Section 6 explores the insights, and concludes this article with a discussion on research limitations and future directions.

2. Literature review and hypotheses development

Business operations is a process of converting resource inputs into outputs and making profits from it. For enterprises, the stragegy of resources allocation is so important that it may directly decide their output performance. This paper mainly discusses the allocation strategy of three types of key resources investment and their impacts on profitability: innovation, marketing, and sustainability. Among them, marketing investment mainly focuses on advertising and personal selling. In the following, this part will review the relevant research in these four aspects in turn, and put forward research hypotheses and the theoretical model based on the situational characteristics of industries that benefit largely from major public health emergencies.

2.1. The impact of innovation on profitability in the industries that massively benefited from major public health emergencies

Innovation is a concept often mentioned in economics and management, which contains organizational innovation, market innovation, process innovation, product innovation, and service innovation, and plays an important role in promoting economic development and social progress (12, 13). Academically, the existing literature has fully studied the relationship between innovation and enterprise performance in general contexts and find that, under most circumstances, innovation, including incremental and fundamental innovation, can promote the improvement of business performance (14, 15). Specifically, for the profitability, which reflects the ability of an enterprise to obtain returns by using its resources, has also been found by many literatures in recent years that it is significantly affected by innovation (16, 17).

However, some scholars hold the opposite perspective towards the relationship between innovation and performance. For example, Link argues that in view of fierce market competition, especially the imitation of competitors, if companies spend high sums of money to innovate, it can sometimes have a negative impact on the future development of the company (18). Kessler holds a similar view, arguing that the impact of enterprise innovation investment on enterprise performance may be affected by external factors such as the company's infrastructure, scale, and industry attributes. Only when external factors are favorable can enterprises obtain more profits in the market through innovative investment (19). In addition to market competition and external environmental factors, the ability of enterprises to use technology is also an important factor. Generally, enterprise innovation investment does not have a direct impact on performance, but needs to be transformed. Only enterprises with strong technology application ability can promote innovation investment in a direction conducive to improving corporate performance (20). Besides, Xu's et al. study also found that the impact of innovation on financial performance is not significant in large enterprises, and it will show a negative impact in small enterprises (21).

In addition, there are also some studies showing that the relationship between firms' innovation input and their financial performance is nonlinear or insignificant. For instance, Lee et al. (22) and Yoo et al. (23) found that the relationship between firm's R&D (Research & Development) expenditure and its ROA (Return on Asset) performance is nonlinear, and influenced by home region orientation and firm lifecycle stages. Koellinger found that despite there is indeed a certain correlation between innovation input, business revenue increase and employment rate improvement, if the enterprise operating revenue is used as the measurement, innovation investment does not necessarily bring more profits to the enterprise, and there is no significant positive correlation between it and enterprise performance (24).

Therefore, as innovation is often long-term and being slow in the process from resource input to output benefit, it may not always lead to better financial performance in the short term. Particularly, if profitability is used as a measuring indicator, the company's R&D revenue may not be able to compensate for R&D costs, resulting in a negative correlation between the two. This negative correlation may become more pronounced in the industries that hugely benefited from suddenly arisen market opportunities, for example, the healthcare industry during COVID-19. This is because if companies ignore market opportunities and continue to focus intensively on R&D, even if they do not consider the risk that the epidemic will end after the development of products related to the treatment of the new crown pneumonia epidemic, there is still the risk that competitors will crowd out market share and reduce innovation revenues and profitability. Thus, we propose the following hypotheses,

Hypothesis 1: In the industries that massively benefited from public health emergencies, innovation would have a negative impact on profitability during the emergency.

2.2. The impact of advertising on profitability in the industries that massively benefited from major public health emergencies

Advertisements are messages paid for by those who send them and are intended to inform or influence people who receive them, as defined by the Advertising Association of the UK (25). From this definition, we can clearly see that advertising costs money, and for enterprises, it requires resource investment. However, in most cases, this investment is worth it and can lead to better financial performance for the business. In recent years, Cici-Karaboga and Sekeroglu found that advertising expense spending has a significant boost to financial performance based on the data of well-known listed companies (10). Mirza et al. also found that not only the current advertising expenditure has a significant role in promoting the financial performance of enterprises, but also the advertising investment in the past period would still play a residual role (26). Niu and Ma's research found that advertising expenditures can promote the positive effects of technology on financial performance, playing a significant moderating role between the two (27).

Nevertheless, some studies have also found that the impact of advertising on corporate performance is not significant or negative. For example, Pourkarimi and Kam found that increasing advertising expenditures alone had no significant effect on market share and sales performance (28). Based on the data from Indian pharmaceutical industry, Pal and Nandy's research also indicate that the impact of advertising on ROA performance is not significant (29). Besides, Xu's et al. research found that for small companies, advertising expenditure can put economic pressure on the business, which may have a negative impact on performance (21). Moreover, the effectiveness of corporate advertising is also affected by advertising content and efficiency (30, 31).

In major public health emergencies, on the one hand, consumers tend to actively increase their search for related products and services due to concerns about their own safety, thereby reducing the marginal utility of the advertising promotion function. On the other hand, major public health events often lead to economic recession, business closures and lower income levels in the affected regions. For enterprises that have benefited from the crisis, if they do not actively assume social responsibility but to advertise their products and services in large quantities, they may easily attract consumer disgust. Therefore, we propose the following hypothesis,

Hypothesis 2: In the industries that massively benefited from public health emergencies, advertising would have a negative impact on profitability during the emergency.

2.3. The impact of personal selling on profitability in the industries that massively benefited from major public health emergencies

In marketing practice, companies often employ multiple marketing methods at the same time, including advertising, personal selling, sales promotion, public relations (32, 33), etc. Among them, personal selling is the use of sales agents to personally deliver messages to target audiences. Despite the complex marketing techniques and online sales, salespeople still play an important role in building relationship with customers and expressing personal care, attention, emotional contact, etc., which are essential in marketing (34, 35). The current literature has amply demonstrated that personal selling has a positive effect on reversing negative public perceptions, promoting product information, and ultimately promoting sales and improving business performance (36–38).

Compared to advertising, personal selling is better targeted and private. In major public health emergencies, beneficiary enterprises can better compensate for the deficiencies of advertising and promote sales through targeted door-to-door marketing. Moreover, during major public health events, consumers often go out less frequently to reduce the risk of infection. Taking the initiative to sell products to the customer's address can also better meet the needs of customers, especially some personalized needs, thereby enhancing the profitability of enterprises. Consequently, we propose the following hypothesis,

Hypothesis 3: In the industries that massively benefited from public health emergencies, personal selling would have a positive impact on profitability during the emergency.

2.4. The impact of sustainability on profitability in the industries that massively benefited from major public health emergencies

With the advancement of industrialization and the attention to environmental and social issues, sustainable development has increasingly become a common goal pursued by countries around the world, and the concept of sustainability has gradually been accepted by all sectors of society. At present, the mainstream view in academia is that sustainability mainly includes three dimensions: environmental sustainability, social sustainability, and governance sustainability (39, 40). In the firm level, sustainability is connected to the extent to which enterprises and their products and service quality meet the environmental, social and governance dimensions, and it has been closely related to the impacts that an enterprise can provide to the society when playing a state role and substituting the function of government from the macro perspective (41, 42). Most existing studies on the relationship between sustainability and corporate performance reveal a positive correlation. For example, Chelawat et al. found that good sustainability performance of enterprises is positively correlated with their financial performance (43). Xie et al. also believe that there is no doubt about the positive correlation between environmental, social and governance sustainability activities and a company's asset returns performance (44). De Lucia et al. found that there is a positive correlation between sustainability practices and financial indicators such as ROA and ROE (Return of Equities) (45). Recently, Zhang et al. found that the enterprises with good sustainability performance will gain a relative competitive advantage during financing, and can obtain funds at a lower financial cost, saving the operating cost of the enterprise, which is conducive to improving the performance of the enterprise (46). Zhang et al. discovered that sustainability may enhance firm performance by alleviating financial constraints, improving external financing, and sending high quality signals to the credit market (47). When the major public health emergencies occur, companies that hugely benefited can also improve their reputation, corporate image, and boost sales and increase profitability by performing with high ratings on sustainability. Hence, we propose the following hypothesis,

Hypothesis 4: In the industries that massively benefited from public health emergencies, sustainability would have a positive impact on profitability during the emergency.

In fact, corporate sustainability not only directly improves profitability, but also influences the relationship between innovation, advertising, personal selling, and profitability. Innovation plays positive role in the long-term development of enterprises, industries and even the economy and society, and is one of the important ways for the sustainable development of enterprises. Companies that emphasize sustainability tend to have a long-term layout and higher requirements for innovation, which is less likely to be significantly reduced by short-term input–output efficiency. During major public health emergencies, companies in industries facing large market opportunities are likely to be detrimental to short-term market competition and financial performance if they continue to focus on innovation and ignore the market. This is more evident in highly sustainable companies. Hence, we propose the following research hypothesis:

Hypothesis 5: In the industries that massively benefited from public health emergencies, sustainability would negatively moderate the relationship of innovation and profitability during the emergency.

Companies with good sustainability not only have positive impact on financial performance, but also usually have the characteristics of good credit quality and strong anti-risk ability (48). Extant research manifests that sustainability would contribute to customer satisfaction, corporate image, and reputation. In the process of marketing, customer cognition and satisfaction of the enterprise may also have an important impact on marketing effect and product sales. Therefore, sustainability may play a positive role in the relationship between advertising, personal selling, and profitability. In the context of major public health emergency, most businesses and residents are facing greater economic pressure. As a company benefiting from major public health emergency, it is necessary to establish a good corporate image to avoid the negative perception that this company is making a fortune during the disaster in the eyes of the public. A good sustainability rating can also promote the consumer recognition of a company and purchasing behavior. Hence, we propose the following research hypotheses:

Hypothesis 6: In the industries that massively benefited from public health emergencies, sustainability would positively moderate the relationship of advertising and profitability during the emergency.

Hypothesis 7: In the industries that massively benefited from public health emergencies, sustainability would positively

moderate the relationship of personal selling and profitability during the emergency.

Many studies have also found that firm age, firm size, and financial leverage have a significant impact on the firm's performance (49, 50), but whether this effect is positive or negative is influenced by many factors and the environment (51-53). Specifically, studies often come to the opposite conclusion when it comes to profitability. In the studies on the age of firms, Ismail et al. found that as companies age, they have more efficient manufacturing capabilities and higher return on investment (54), but Loderer's et al. research found that the profitability of a business declines with age (55). In the studies about firm size, on one hand, Kuncová et al. (56), Ibhagui & Olokoyo (57) and Mubeen et al. (58) found that large firms outperformed small firms in profitability, and firm size can positively moderate the relationship between factors such as financial leverage, product market competition and firm investment performance. On the other hand, Dhawan (59) and Fan's et al. (60) studies found that firm size is negatively correlated with company performance. Similarly, there may also be a positive or negative relationship between financial leverage and profitability. Fosu (61) and Iqbal & Javed's (62) studies found that financial leverage's impacts on profitability is positive, while Shahzad et al. (63) and Mathur's et al. (64) studies obtained the opposite results. Considering that these factors may also have significant impacts on profitability, we included them as control variables in the empirical model for analysis in this study. Finally, the research model of this study is shown in Figure 1.

3. Materials and methods

3.1. Samples and data collection

We calculated the specific ROA data and their changes of listed companies in various industries of China's A-share market between 2019 (the year before the outbreak) and 2021 (the second year after the outbreak of the new crown epidemic, latest data) first to find the industries that massively benefited from COVID-19. Through the statistics of a total of 4,929 companies across 11 industries, we find that the healthcare industry ranks first with an ROA growth rate of 68.86%, the materials industry ranks second with an ROA growth rate of 14.96%, and the average growth rate of 11 industries is -5.46%. Consequently, the healthcare industry is the only one that has gained high growth during the pandemic, and we conducted empirical research on this industry.

Healthcare industry is an aggregation and integration of sectors within the economic system that provides goods and services to treat patients with curative, preventive, rehabilitative, and palliative care, which includes the generation and commercialization of goods and services for maintaining and re-establishing health (65). In China, the healthcare industry is mainly made up of two main categories of companies. One category is healthcare equipment and service enterprises, including healthcare equipment, healthcare supplies, healthcare providers and services, healthcare technology, etc. The other category is pharmaceutical, biotechnology and bioscience enterprises, including traditional Chinese medicine, modern medicine, biotechnology, bioscience tools and services, etc. We used the data of China healthcare industry in this study. According to Wind database, as of December 31, 2021, there were 484 listed companies. Wind is a comprehensive database company of China, whose data are widely used in academic researches (66–68). Finally, after excluding samples containing missing values, 364 companies were selected. From the average level, the ROA of 364 companies selected by this study is 9.22%, which is basically equal to the mean value of 9.02% in the whole healthcare industry, indicating that the sample of this study has a good representativeness. All data used in this article is secondary data from publicly available database and can be openly accessed and verified.

3.2. Variables

The dependent variable in this study is profitability, which reflects the ability of a firm to generate revenue from resources. In financial analysis, there are many indicators to measure profitability, including return on assets, return on net assets, operating profit margin, cost and expense margin, surplus cash protection multiple, etc. Among them, return on assets is the most widely used variable, which reflects the ability of enterprises to use all assets to obtain profits. Compared with net profit, operating income, sales profit margin and other indicators, ROA can more intuitively reflect the input–output efficiency of enterprise resources.

In this study, we mainly analyzed four explanatory variables. The first is innovation. Most of the empirical studies draw on the practices of Miller and Le Breton-Miller (69) and measure the ratio of annual R&D expenditures to annual sales. Using this approach, we measured innovation with the ratio of R&D expenditure to sales revenue in 2021. We normalized the variable using the following formula:

Normalized Innovation_{*i*} = (Innovation_{*i*}-mean(innovation))/(Max (Innovation)-Min (Innovation)) (i = 1, 2, 3...).

The second explanatory variable is advertising. For listed companies, advertising is an important tool in corporate marketing, and its related expenses are usually clearly listed in the financial report. Referring to the calculation method of innovation, we calculate advertising using the ratio of advertising cost to product sales revenue. During data analysis, this variable is also normalized so that its value ranges from 0–1.

The third explanatory variable is personal selling. Generally, the expenditure of enterprises on personal selling has a significant positive correlation with the proportion of salesmen. Considering that personal selling expenditure data is not directly disclosed in the company's annual report, we use the ratio between the number of sales personnel and the total number of employees as a proxy variable. Likewise, this variable is normalized during data analysis so that its value ranges from 0-1.

The fourth explanatory variable in this study is sustainability. In general, there are two main methods for measuring sustainability in the existing literature. One is to directly quote the assessment values of third-party institutions, which is becoming more reliable and recognized as sustainability becomes more important in society and the maturity of assessment methods, data collection scope, etc. (70, 71). Another method is to design empirical indicators by the researchers themselves according to the definition of sustainability. By strictly following the dimensions and content of the concept and collecting data through questionnaire and other methods, it can cover



small and medium-sized enterprises and non-listed companies that are rarely involved in third-party institutions (12, 72). Since the samples in this study are all publicly listed companies, we took the first measurement method, which directly references Wind's sustainability rating data. Specifically, based on the division of Wind, we divide sustainability into three dimensions: environmental sustainability, social sustainability, and governance sustainability.

Considering that a firm's age, size, and risk-taking can also have an impact on its profitability, we included them in our study as control variables (51–53). Among them, firm age is measured by the number of days from the date of firm registration to January 1, 2021, and we performed logarithmic processing on the original data. In terms of firm size, we control two main variables: asset value and number of employees. The corresponding data comes directly from the company's annual report and is logarithmic processed before analysis. The last control variable is debt to net worth ratio, which measures the debt risk afford by shareholders and is calculated as the ratio between total liabilities and shareholders' equity. The descriptive statistics of the variables are shown in Table 1.

3.3. Research methodology

This study mainly focuses on the impact of innovation, advertising, personal selling, and sustainability on firm profitability in the industries that massively benefited from major public healthcare emergencies. We first conducted statistics on the changes in the return on assets of various industries in China's A-shares since the outbreak of the new crown pneumonia epidemic, and found that the healthcare industry is one of the three industries that have achieved growth in 11 major industries, and the growth rate is far higher than that of other industries. Therefore, we use the healthcare industry as sample source and hierarchical regression (ordinary least squares regression), moderating effect test and robust test as statistical methods. Based on the theoretical review and research hypotheses, the empirical analysis of this study mainly includes three parts: (1) to test the impact of innovation, advertising, personal selling and sustainability on corporate profitability; (2) to test the moderating effect of sustainability and its sub-items on the relationship between innovation, advertising, personal selling and profitability; (3) Robustness tests are carried out by increasing the measurement range of sustainability variable, changing data source and parameter estimation method, as well as quantile regression.

4. Results

4.1. The impact of innovation, advertising, personal selling, and sustainability on profitability

We used the ordinary least squares estimation method and hierarchical regression models to analyze the impact of innovation, advertising, personal selling and sustainability, and Table 2 shows the detail regression results. Specifically, Model 1 is a model with

Variables	1	2	3	4	5	5.1	5.2	5.3	6	7	8	9
1. Profitability	1.00	-	-	-	-	-	-	-				
2. Innovation	-0.34***	1.00	-	-	-	_	_	_				
3. Advertising	-0.18***	0.03	1.00	-	-	-	-	-				
4. Personal Selling	0.09	0.01	0.18***	1.00								
5. Sustainability	0.21***	0.02	-0.14***	0.01	1.00	-	-	_				
5.1 Environmental Sustainability	-0.03	-0.08	0.05	-0.06	0.65***	1.00						
5.2 Social Sustainability	0.23***	0.08	-0.14***	0.03	0.85***	0.36***	1.00					
5.3 Governance Sustainability	0.21***	0.04	-0.14***	0.04	0.68***	0.23***	0.36***	1.00				
6. Ln (Firm Age)	-0.10**	-0.20***	0.01	0.01	-0.01	0.12**	-0.10*	-0.01	1.00			
7. Ln (Asset Size)	0.08	-0.10*	-0.16***	0.05	0.25***	0.39***	0.08	0.20***	0.20***	1.00		
8. Ln (Total Number of Employees)	0.05	-0.10*	-0.17***	0.09*	0.25***	0.40***	0.08	0.16***	0.26***	0.84***	1.00	
9. Debt to Net Worth Ratio	-0.10*	-0.20***	-0.004	0.08	-0.10*	0.049	-0.12**	-0.06	0.15***	0.05	0.08	1.00
Mean	9.02	0.02	0.15	0.29	0.42	0.24	0.44	0.66	8.89	12.87	7.47	68.16
Minimum	-55.31	0	0	0	0	0	0	0	6.82	10.46	4.91	201.08
Maximum	113.06	1	1	1	1	1	1	1	9.68	16.61	10.76	-708.22
Std. Deviation	13.35	0.06	0.20	0.21	0.16	0.22	0.21	0.13	0.34	1.05	1.06	3298.76
Valid N	364	364	364	364	364	364	364	364	364	364	364	364

TABLE 1 The descriptive statistics of variables.

***p<0.01, **p<0.05, and *p<0.10, two tailed test.

only control variables, and Model 2 is the model containing control variables, innovation, advertising, personal selling, and sustainability; Model 3 breaks down sustainability of Model 2 into environmental sustainability, social sustainability, and governance sustainability.

It can be seen from Model 2 that, on one hand, innovation and advertising investment have a significant negative impact on profitability, which proves the H1 and H2. On the other hand, personal selling and sustainability play a significant positive role in profitability, which verifies the correctness of H3 and H4. Model 3 once again confirms the negative impact of innovation and advertising on profitability, and the positive impact of personal selling. For the specific three dimensions of sustainability, Model 3 manifests that environmental sustainability has a negative impact on profitability, while social sustainability and governance sustainability have a positive impact on profitability. These results conflict with the positive impact of innovation and advertising on firm performance in the general context, but validates the hypotheses H1 and H2. For the four control variables, firm age and debt to net worth ratio are proved to have negative impact on profitability according to the Table 2, which is consistent with the studies of Shahzad et al. (63), Loderer et al. (55), Mathur et al. (64) and so forth. This suggests that both older and more risky companies perform relatively poorer in terms of profitability. Besides, the two variables measuring firm size including asset size and employee quantity are found insignificant to profitability.

4.2. The moderating effect of sustainability

In Section 4.1, we confirmed the role of sustainability in boosting corporate profitability. As a matter of fact, sustainability may also

moderate the relationship between innovation, advertising personal selling and profitability. In this section, we analyze whether there are moderating effects by incorporating the interaction of sustainability with innovation, advertising, and personal selling into the regression models. Particularly, Model 4 is the regression of profitability on innovation, advertising, personal selling, sustainability, the product of sustainability and innovation, the product of sustainability and advertising, the product of sustainability and personal selling, and four control variables. Model 5 builds on Model 4 and further analyzes the moderating effect of sustainability on innovation, distinguishing between environmental sustainability, social sustainability, and governance sustainability. Results are shown in Table 3.

In model 4, when the interaction items of sustainability and innovation, advertising and personal selling are added, only the coefficient of the interaction item of innovation and sustainability is significantly less than 0, and the regression coefficient of the interaction items of advertising and promotion and sustainability are not significant. This indicates that sustainability has a negative moderating effect on the relationship between innovation and profitability, but it does not significantly moderate the relationship between advertising, personal selling, and profitability. Therefore, the H5 is proved to be right while H6 and H7 are not supported by our data. Furthermore, Model 5 specifically demonstrates the moderating effect of the three dimensions of sustainability on the relationship between innovation and profitability. Judging from the statistical results, the moderating effect of environmental sustainability and governance sustainability on the relationship between innovation and profitability is not significant, while social sustainability has a significant negative moderating effect. That is, for companies with high social sustainability, their profitability will gradually diminish as investment in innovation increases. But for companies with low TABLE 2 OLS and hierarchical regression results.

Dependent variable: Profitability	Model 1	Model 2	Model 3
Constant	27.908	58.008***	29.922
Independent Variables:			
Innovation		-86.603***	-91.676***
Advertising		-11.567***	-7.837**
Personal Selling		8.357***	5.906**
Sustainability		14.513***	
Environmental Sustainability			-10.840***
Social Sustainability			14.247***
Governance Sustainability			14.453***
Control Variables:			
Ln (Firm Age)	-4.307**	-5.849***	-5.164***
Ln (Asset Size)	1.799	1.047	1.273
Ln (Total Number of Employees)	-0.454	-1.164	-0.337
Debt to Net Worth Ratio	-0.006	-0.010***	-0.010***
Model-fitting metrics:			
R ²	0.030	0.2451	0.2902
Adjusted R ²	0.019	0.2281	0.2701
F value	2.75**	14.41***	14.43***
Observations	364	364	364

***p<0.01, **p<0.05 two tailed test.

social sustainability, increased levels of innovation can still boost profitability. This indicates that the negative moderating effect of sustainability on the relationship between innovation and profitability is mainly caused by the dimension of social sustainability, as can be seen in the Figure 2.

Finally, we collate the validation result of the 7 hypotheses presented in Section 2, as shown in the Table 4.

4.3. Robustness check

4.3.1. Increasing the measurement range of sustainability

In the regression analysis above, we use internationally widely used concept of sustainability to analyze its impact on profitability. This indicator mainly includes three dimensions of environmental, social and governance factors. In recent years, public opinion, regulatory penalties, and legal proceedings have attracted more and more attention from consumers and the external public, and have significantly affected corporate image and business performance. Therefore, Wind added the dimension of dispute events to the traditional three dimensions of sustainability, and calculated the comprehensive score of sustainability. We use the comprehensive sustainability score calculated from these four dimensions to replace the traditional sustainability score, and then observe the impact of each related factor on profitability. Results are shown in Table 5.

Model 6 is the regression of profitability on innovation, advertising, personal selling, comprehensive sustainability and four control variables, from which we can see that innovation and advertising' s impact is negative and personal selling and sustainability's impact is positive. Model 7 adds three additional interactive items of sustainability and innovation, advertising, and personal selling to Model 6. The result shows that only the regression coefficient of the interaction term between sustainability and innovation is negatively significant, while the regression coefficients of the other interaction items are not significant. Besides, for the four control variables, the significance of coefficients is also in accordance with original models. Consequently, after replacing the traditional sustainability score with the comprehensive sustainability score, the results are still consistent, which proves the robustness of this study.

4.3.2. Changing the data source and parameter estimation method

In this part, we further test the robustness by changing the data source of sustainability variable and the estimation method. Sustainability is a key independent variable in this study, and the accuracy and objectivity of its measurement directly affect the research conclusion. In the study above, we used Wind data to measure sustainability. In China, Sino-Securities ESG Index is another highly recognized sustainability score data, which comprehensively considers the internal management system, business objectives, green products, external certification, violations and other factors, and the basic data used is mainly based on the public disclosure data of listed companies, and integrates auxiliary information such as social responsibility reports, sustainable development reports, national regulatory announcements, news media data, etc. We convert the original nine grades of C/CC/ CCC/B/BB/BBB/A/AA/AAA into numerical values from 1 to 9, respectively, to simplify the quantitative analysis. To be specific, C

Dependent varial	ole: Profitability	Coef.	Std. Err.	t	P > t
Model 4 ($R^2 = 0.268$,	Constant	60.771***	18.223	3.33	0.001
F=11.72, Sig<0.001, N=364)	Innovation	99.416*	58.089	1.71	0.088
	Advertising	-11.302	10.182	-1.11	0.268
	Personal Selling	12.403*	7.499	1.65	0.099
	Sustainability	27.032***	7.438	3.63	0.000
	Sustainability *Innovation	-537.538***	165.363	-3.25	0.001
	Sustainability *Advertising	-1.359	24.974	-0.05	0.957
	Sustainability *Personal Selling	-8.400	17.266	-0.49	0.627
	Ln (Firm Age)	-5.876***	1.897	-3.10	0.002
	Ln (Asset Size)	1.128	1.095	1.03	0.304
	Ln (Total Number of Employees)	-1.542	1.117	-1.38	0.168
	Debt to Net Worth Ratio	-0.009***	0.003	-2.92	0.004
Model 5 ($R^2 = 0.312$,	Constant	28.541	18.547	1.54	0.125
F=12.18, Sig<0.001,	Innovation	126.008	150.135	0.84	0.402
N=364)	Advertising	-8.059**	3.275	-2.46	0.014
	Personal Selling	6.616**	2.908	2.27	0.024
	Environmental Sustainability	-7.481*	3.945	-1.90	0.059
	Social Sustainability	19.480***	3.976	4.90	0.000
	Governance Sustainability	16.852***	5.864	2.87	0.004
	Environmental Sustainability*Innovation	-215.977	144.415	-1.50	0.136
	Social Sustainability*Innovation	-281.360**	141.499	-1.99	0.048
	Governance Sustainability*Innovation	-120.217	224.370	-0.54	0.592
	Ln (Firm Age)	-5.382***	1.883	-2.86	0.005
	Ln (Asset Size)	1.413	1.078	1.31	0.191
	Ln (Total Number of Employees)	-0.677	1.097	-0.62	0.538
	Debt to Net Worth Ratio	-0.009***	0.003	-2.88	0.004

TABLE 3 Sustainability and its sub-items' moderating effects.

****p* < 0.01, ***p* < 0.05, and **p* < 0.10, two tailed test.

has the lowest grade and is assigned the value of 1. AAA has the highest rating and is assigned the value of 9. Among the other grades, CC grade is assigned the value of 2, CCC grade is assigned the value of 3, B grade is assigned the value of 4, BB grade is assigned the value of 5, BBB grade is assigned a value of 6, A grade is assigned the value of 7, and AA grade is assigned the value of 8. We use the robust or sandwich estimator of variance method to run the regression models to further eliminate some types of misspecifications, which uses (sigma-hat_j)^2 = {n/(n-k)}(u_{-j})^2 as an estimate of the variance of the *j*th observation, where u_{-j} is the calculated residual and n/(n-k) is included to improve the overall estimate's small-sample properties, and the results are shown in Table 6.

As with the models in the robustness test of 4.3.1, Model 8 is the regression of profitability on innovation, advertising, personal selling, Sino-Securities sustainability and four control variables, from which we can see that innovation and advertising's impact is still negative and personal selling and sustainability's impact is still positive. Model 9 adds three additional interactive items and the result again manifests that only the regression coefficient of the interaction item between

sustainability and innovation is negatively significant, while the regression coefficients of the other interaction items are not significant. In addition, for the four control variables, the significance of coefficients is also in accordance with original models and the robustness test in Section 4.3.1. Therefore, we can draw the conclusion that when the Wind sustainability score is replaced by the Sino-Securities sustainability index, the results are still consistent, which once again proves the strong robustness of this research.

4.3.3. Quantile regression results with bootstrap

In order to ensure the robustness of this study and observe the impact of innovation, advertising, personal selling, and sustainability on profitability at different profitability levels, we further analyzed the data using quantile regression and bootstrap methods. The number of bootstraps is set to 100, and the median regression results are shown in Table 7.

As can be seen from Table 7, the impact of innovation and advertising on profitability remains significantly negative in terms of median, while the influence of personal selling and sustainability on profitability remains significantly positive. This is consistent with the results of the OLS regression above. We further demonstrate quantile regression results in Figure 3 of the impact of innovation, advertising, personal selling, and sustainability on profitability. At all quantile levels, the regression coefficients of innovation and advertising are both less than 0, while the regression coefficients of personal selling and sustainability are both greater than 0. This again shows that the results of this study have strong robustness.

5. Discussion

5.1. The resource allocation of innovation and marketing and firm profitability

The epidemic of the new crown pneumonia has had a serious impact on the world economy (73–75). However, the healthcare industry has certainly been a beneficiary industry amid this pandemic and brought with huge market demand. How to seize this opportunity and make the right innovation and marketing strategies is an important issue for enterprises.

TABLE 4 Hypotheses test result.

Hypothesis	Supported or Not supported
H1	YES
H2	YES
Н3	YES
H4	YES
H5	YES
H6	NO
H7	NO



The role of innovation in business performance and economic development has been widely recognized (14, 15, 76). However, from investing resources in research and development to bringing new

TABLE 5 Robust test results by increasing the measurement range of sustainability.

Dependent variable: Profitability	Model 6	Model 7
Constant	48.793***	44.630**
Independent Variables:		
Innovation	-86.443***	117.701*
Advertising	-11.390***	-13.881
Personal Selling	8.217***	12.084
Comprehensive Sustainability	15.174***	26.082***
Interaction Items:		
Comprehensive Sustainability*Innovation		-536.494***
Comprehensive Sustainability*Advertising		5.540
Comprehensive Sustainability*Personal Selling		-8.087
Control Variables:		
Ln (Firm Age)	-5.702***	-5.530***
Ln (Asset Size)	1.197	1.260
Ln (Total Number of Employees)	-1.254	-1.557
Debt to Net Worth Ratio	-0.009***	-0.009***
Model-fitting metrics:		
R^2	0.2479	0.2664
Adjusted R ²	0.2309	0.2435
<i>F</i> value	14.63***	11.62***
Observations	364	364

*****p* < 0.01, ****p* < 0.05, and **p* < 0.10, two tailed test.



FIGURE 2

The moderating effect of overall sustainability and social sustainability.

10

technologies to market, it is often a long-term process, which determines that innovation should be a long-term behavior (77). On the other hand, as Link (18) and Kessler (19) have argued, market competition and the external environment are also important factors that should be considered in the innovation decisions of enterprises. When faced with a sudden huge market opportunity, if the

TABLE 6 Robust test results by changing data source and estimation method.

Dependent variable: Profitability	Model 8	Model 9
Constant	58.078***	55.281***
Independent Variables:		
Innovation	-80.223***	-2.807
Advertising	-10.636***	-18.334*
Personal Selling	7.079**	19.916
Sino-Securities Sustainability	8.031**	14.017***
Interaction Items:		
Sino-Securities Sustainability*Innovation		-169.651**
Sino-Securities Sustainability*Advertising		13.446
Sino-Securities Sustainability*Personal Selling		-21.367
Control Variables:		
Ln (Firm Age)	-7.389***	-7.591***
Ln (Asset Size)	1.610	1.988
Ln (Total Number of Employees)	-0.922	-1.368
Debt to Net Worth Ratio	-0.010**	-0.008***
Model-fitting metrics:		
R ²	0.2327	0.2447
F value	19.14***	82.31***
Observations	350	350

***p<0.01, **p<0.05, and *p<0.10, two tailed test.

TABLE 7 Median regression results.

competitors of a firm have obtained higher sales performance and market share through the adjustment of resource allocation, and this firm still insists to focus on R&D, it will inevitably be detrimental to its financial performance and competitiveness. Particularly, for the industries that have benefited greatly from major public health emergencies, the market demand has increased significantly, and companies are likely to be outperformed by competitors if they fail to seize the opportunity. At the level of firm practice, empirical data also confirm this negative relationship between innovation and profitability. For enterprises that benefit greatly from major public health emergencies, increasing their ability to respond to market dynamics and appropriately adjusting the investment of innovation resources according to the market environment will be more conducive to improving their current performance. However, in the long run, we still believe that companies should maintain and increase investment in innovation to improve their core competitiveness.

Compared to the internal orientation character of innovation, advertising and personal selling are directly market-oriented and have more advantages in responding to the market opportunities that suddenly arise in the market. Of the two, advertising of healthcare industry can not only enable consumers to fully understand product and service information and enhance health awareness, but also facilitate the communication and concordance between customers and doctors (78, 79). If the advertising strategy is well devised and deployed, it may foster provider-patient engagement initiatives and offer healthcare providers opportunities to dramatically improve their sales and performance by successfully engaging current and prospective patients and hastening exchange (80, 81). However, advertising in the healthcare industry is sometimes not objective and can be misleading, creating unrealistic expectations for consumers (82). Some healthcare industry consumers may feel that while commercial health advertising is helpful, it is also confusing, with many participants also holding mistaken beliefs around other elements of the commercial healthcare advertising (83). The chaos of advertising in the healthcare industry, as well as consumer

Profitability	Coef.	Std. Err	t	p > t	[95% Con	f. Interval]		
Constant	61.421	12.956	4.74	0.000	35.941	86.901		
Independent Variables:								
Innovation	-81.662	30.384	-2.69	0.008	-141.418	-21.906		
Advertising	-7.117	2.308	-3.08	0.002	-11.657	-2.578		
Personal Selling	6.494	2.144	3.03	0.003	2.278	10.710		
Sustainability	9.898	3.305	2.99	0.003	3.397	16.398		
Control Variables:								
Ln (Firm Age)	-5.167	1.459	-3.54	0.000	-8.036	-2.297		
Ln (Asset Size)	-0.699	0.662	-1.06	0.292	-2.002	0.603		
Ln (Total Number of Employees)	0.403	0.887	0.45	0.650	-1.342	2.148		
Debt to Net Worth Ratio	-1.687	1.049	-1.61	0.109	-3.750	0.376		
Model-fitting metrics:								
Pseudo R ²	0.1472							
Observations	364							



misunderstanding of advertising for healthcare products, has dampened the positive effect of advertising on profitability. Besides, in order to deal with this problem, various measures have been taken by the regulatory agencies all over the world to effectively control and curb the misleading or false claims through strict regulations, which also increase the cost of advertising activities (84, 85). Moreover, in major public health emergencies, the demand for healthcare products is increased significantly, and consumers' active product search behavior is also increased correspondingly, which may reduce the effectiveness of advertising to a certain extent. Ultimately, as shown in this paper, these factors make the output of advertising lower than the cost of input, and thus has a negative impact on profitability. For enterprises that have benefited greatly from major public health emergencies, paying attention to the correct promotion of advertising, and keeping a low profile, and appropriately reducing advertising expenditure will be more conducive to improving the efficiency and return on investment of resources.

Although the audience of advertising is very broad, it is relatively weak in terms of public relations with focused customers. To supplement this deficiency, personal selling is another important marketing tool often deployed in the healthcare industry (86). Healthcare products and services have a certain degree of professionalism, whose main target customers are hospitals, pharmacies, health care institutions, local governments, and other organizations with large procurement batches, which determines that personal selling would play an important role in the marketing mix (87–89). Especially in China, affected by the social and cultural factors represented by "GUANXI" (also called Personal Connections), if sales personnel could maintain good relationships with important customers such as local governments, hospitals, pharmacies, and health management companies, it will also have a positive role in promoting product sales (90–92). In fact, the sale skills of sales professionals in the healthcare industry and the management of GUANXI with customers are common in countries around the world, except some differences in ethical or legal norms (93, 94). Much of the literature in recent years also proves that the personal characteristics and abilities of pharmaceutical sales representatives have a significant impact on market share and sales performance (36, 37). Therefore, for companies benefiting from major public health emergencies, the personal selling targeted at focused customers would outperform large-scale advertising in improving profitability.

In a word, the market and consumer demand for healthcare products have increased significantly during major public health emergencies, and increasing resource investment and training in personal selling is more conducive to fully grasping market opportunities, thereby improving the profitability of enterprises. Consequently, it is not very unexpected that the impact of investing in personal selling on the profitability of enterprises in the current period is positive, while the investment in innovation and advertising would cause negative impacts.

5.2. The impact of sustainability and its moderating role

The major public health emergency is a kind of social disaster which would lead to serious losses of country, firms, and individuals. In this context, although some industries and companies may profit from the increase of market demand, the public still expects them to take the initiative to assume social responsibility rather than make lots of money during the disaster. In this context, enterprises increase sustainability investment and actively assume social responsibility are also a kind of marketing and publicity for themselves (95, 96). Especially when the industry is mixed with fishes and dragons and the public has doubts about the reputation of the industry, the investment of sustainability of healthcare companies will be conducive to the improvement of brand influence and business performance. From the statistical results of this study, the impact of sustainability on corporate profitability is twofold. On one hand, it can directly promote the improvement of corporate profitability. On the other hand, it also significantly moderates the relationship between innovation and profitability.

In hierarchical regression models, sustainability has a significant positive impact on corporate profitability, which verifies our analysis above. Specifically, among the three dimensions of sustainability, the impact of environmental sustainability on profitability is negative, while the impact of social sustainability and governance sustainability on profitability is positive. This suggests that the three dimensions of sustainability are not aligned in their directions. While the pursuit of sustainability can increase profitability, investment in environmental sustainability is still negative if viewed solely from the perspective of improving profitability. This result is also simple to understand: social sustainability and governance sustainability are oriented to the internal and external governance structure of the enterprise, and have a direct relationship with operation, products, market, and customers, so that it is easier to improve profitability by investing in these areas. But environmental sustainability is not the case. At least for now, the output of enterprises' investment in environmental system and management system construction, resource conservation and protection, reduction of harmful emissions, green building and green finance is still lower than cost. How to make the investment of environmental sustainability to have a greater support for profitability remains one of the key issues for the healthcare industry and companies. In fact, regardless of whether a company's investment in environmental sustainability results in a favorable return on investment, the image of social sustainability and governance sustainability is even more important for companies that profit a lot from a major public health emergency. Increased investment in social sustainability and governance sustainability is better for companies to make greater profits in a disaster environment.

We also find that sustainability may moderate the negative impact of innovation on profitability during the moderating effect analysis, which is somewhat unexpected while makes sense. Sustainability has been one of the key goals pursued by many companies since this century, and it emphasizes long-term development by conserving resources, protecting the environment, and establishing good internal and external relationships. The more sustainable companies are, the more emphasis is placed on long-term development through innovation, rather than reducing investment in innovation for reasons such as seizing current market opportunities and the short-term output efficiency of innovation. Ultimately, this leads to a more negative impact of innovation on current profitability in highly sustainable companies. Further analysis manifests that this moderating effect of sustainability is mainly caused by social sustainability, which primarily measures the performance of enterprises in terms of employment relations, occupational health, production safety, product quality, customer privacy protection, and community public welfare investment and reflects the responsibility and friendliness of enterprises at the social level. Companies with high social sustainability have a greater negative impact of their innovations on profitability than firms with low social sustainability as their innovation inputs are more long-term and less market-sensitive. In contrast, environmental sustainability and governance sustainability do not have a significant impact on the relationship between innovation and profitability, which means that the negative impact of corporate innovation investment on profitability will not change depending on environmental sustainability and governance sustainability. Furthermore, the level of sustainability also will not have a significant moderating effect on the profitability impact of advertising and personal selling.

In summary, for enterprises in the industries that profit a lot from major public health emergencies, it is beneficial for them to better seize market opportunities and improve business performance by appropriately tilting resource allocation towards sustainability. Businesses may consider shifting resources from innovation to social sustainability and governance sustainability to improve current profitability, which is strikingly different from the general context that emphasizes increasing profitability through innovation, marketing, and sustainability investments together.

6. Conclusion

The COVID-19 has affected many firm characteristics including firm performance, firm value, governance structure, dividend, liquidity, and leverage level (97, 98). Therefore, it is interesting to study how companies respond and the effects of these countermeasures. Although most industries will suffer losses during major public health emergencies, public healthrelated industries have ushered in lots of market opportunities. This article conducted research on companies in China's healthcare industry during the coronavirus pandemic and found some interesting conclusions in the resource allocation strategies, which are not only consistent with but also supplementary to the previous researches on innovation, marketing, sustainability, and financial performance in general context. On one hand, we found that personal selling, social sustainability, and governance sustainability have a significant positive impact on profitability, just as the situation of many industries in the general context. On the other hand, the conclusion that environmental sustainability, innovation, and advertising have been shown to have a negative impact on profitability is worth pondering. For companies that benefit greatly from major public health emergencies, how to improve the resource efficiency of innovation and advertising tools, and balance environmental protection and financial performance remains an important issue to be addressed by the firms, industrial organizations, and government. Especially for the enterprises with high social sustainability, mechanisms should be put in place to help them reverse the negative impact of innovation on profitability. Finally, in general, for companies in industries that have benefited greatly from major public health emergencies, we believe that a shift in resource allocation from innovation, advertising, and environmental sustainability to personal selling, social sustainability, and governance

sustainability will be more conducive to improving profitability. However, for public health regulatory authorities, it is necessary to strengthen the supervision of sales representatives of health care enterprises, hospitals, public health organizations, etc., and appropriately subsidize the innovation of enterprises to enhance their innovation motivation.

There are two main limitations in this study. First, in terms of sample selection, we used the data of listed companies of China's healthcare industry, which does not include the small and mediumsized enterprises that unlisted on the stock market. Due to the limited sample size and scope, whether the result is suitable for other countries and industries also needs to be further verified. Moreover, like other empirical studies based on secondary data, our models may also suffer endogenous problems caused by the omission of potential explanatory variables and the correlation between residuals and independent variables. Second, this paper is a preliminary study about the industries that massively benefited from major public health emergencies and just focuses on the resource allocation strategy of innovation, marketing, and sustainability, while lacks of deep research in the specific mechanism and processes of innovation, advertising, and personal selling on profitability. We hope that in the future, more scholars and studies may pay attention to these relevant industries and corporate behavior in this special context and the research can focus on at least two directions. One is to study in-depth about the impact mechanism of these enterprises' resource allocation and other competition strategies on performance in the context of major public health emergencies. The second is to discuss about how these industries and enterprises that have benefited massively from the major public health emergencies can achieve long-term sustainable development when the disaster ended.

Data availability statement

Publicly available datasets were analyzed in this study. This data can be found here: https://www.wind.com.cn/en/.

References

1. Hopenhayn HA. Firms, misallocation, and aggregate productivity: a review. Annu Rev Econom. (2014) 6:735–70. doi: 10.1146/annurev-economics-082912-110223

2. Arias M, Saavedra R, Marques MR, Munoz-Gama J, Sepúlveda M. Human resource allocation in business process management and process mining: a systematic mapping study. *Manag Decis.* (2018) 56:376–405. doi: 10.1108/MD-05-2017-0476

3. Fu L, Liao S, Liu Z, Lu F. An investigation of resource allocation mechanism for exploration and exploitation under limited resource. *IEEE Trans Eng Manag.* (2019) 68:1802–12. doi: 10.1109/TEM.2019.2938480

4. Bai T, Liesch P. Organizational goals and resource allocation to overseas foreign direct investment. J World Bus. (2022) 57:101308. doi: 10.1016/j.jwb.2022.101308

5. Symeonidou N, Leiponen A, Autio E, Bruneel J. The origins of capabilities: resource allocation strategies, capability development, and the performance of new firms. *J Bus Ventur.* (2022) 37:106208. doi: 10.1016/j.jbusvent.2022.106208

 Arvidsson S, Dumay J. Corporate ESG reporting quantity, quality and performance: where to now for environmental policy and practice? *Bus Strategy Environ*. (2022) 31:1091–110. doi: 10.1002/bse.2937

7. Broadstock DC, Chan K, Cheng LT, Wang X. The role of ESG performance during times of financial crisis: evidence from COVID-19 in China. *Financ Res Lett.* (2021) 38:101716. doi: 10.1016/j.frl.2020.101716

8. Farza K, Ftiti Z, Hlioui Z, Louhichi W, Omri A. Does it pay to go green? The environmental innovation effect on corporate financial performance. *J Environ Manag.* (2021) 300:113695. doi: 10.1016/j.jenvman.2021.113695

Author contributions

GX and CZ: conceptualization, project administration, and funding acquisition. GX: methodology and software. GX and YZ: validation. GX, CZ, QF, and XT: writing – original draft preparation. QF, XT, and YZ: writing – review and editing. QF and XT: supervision. All authors contributed to the article and approved the submitted version.

Funding

This research was funded by the Special fund for basic scientific research of Central universities (2021QN1054).

Acknowledgments

We appreciate the comments of the reviewers.

Conflict of interest

YZ was employed by Chaobo Medical Materials Farming Cooperatives of Yangcheng.

The remaining authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

Publisher's note

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

9. Aftab J, Veneziani M, Sarwar H, Ishaq MI. Entrepreneurial orientation, entrepreneurial competencies, innovation, and performances in smes of Pakistan: moderating role of social ties. *Business Ethics Environ Responsibility*. (2022) 31:419–37. doi: 10.1111/beer.12415

10. Cici-Karaboga EN, Sekeroglu G. The mediating role of advertising expenditures in the relationship of brand value and financial performance. *Sosyoekonom.* (2021) 29:235–50. doi: 10.17233/sosyoekonomi.2021.03.12

11. Rahman M, Rodríguez Serrano MÁ, Hughes M. Does advertising productivity affect organizational performance? Impact of market conditions. *Brit J Manage*. (2021) 32:1359–83. doi: 10.1111/1467-8551.12432

12. Staub S, Kaynak R, Gok T. What affects sustainability and innovation—hard or soft corporate identity? *Technol Forecast Soc Change*. (2016) 102:72–9. doi: 10.1016/j. techfore.2015.06.033

13. Forcadell FJ, Aracil E. Can multinational companies foster institutional change and sustainable development in emerging countries? A case study. *Business Strat Dev.* (2019) 2:91–105. doi: 10.1002/bsd2.45

14. Sempere-Ripoll F, Estelles-Miguel S, Rojas-Alvarado R, Hervas-Oliver J. Does technological innovation drive corporate sustainability? Empirical evidence for the European financial industry in catching-up and central and Eastern Europe countries. *Sustainability.* (2020) 12:2261. doi: 10.3390/su12062261

15. Huang SY, Chiu A, Lin C, Chen T. The relationship between corporate innovation and performance. *Total Qual Manag Bus Excell.* (2018) 29:441–52. doi: 10.1080/14783363.2016.1209406

16. Canh NT, Liem NT, Thu PA, Khuong NV. The impact of innovation on the firm performance and corporate social responsibility of vietnamese manufacturing firms. *Sustainability.* (2019) 11:3666. doi: 10.3390/su11133666

17. Xie X, Huo J, Zou H. Green process innovation, green product innovation, and corporate financial performance: a content analysis method. *J Bus Res.* (2019) 101:697–706. doi: 10.1016/j.jbusres.2019.01.010

18. Amato L. Research and development activity in u. *S Manufact Southern Econ J.* (1982) 48:1108–10. doi: 10.2307/1058292

19. Kessler EH. Leveraging ER&D processes: a knowledge-based view. *Technovation*. (2003) 23:905–15. doi: 10.1016/s0166-4972(03)00108-1

20. Cuneo P, Mairesse J. Productivity and R&D at the firm level in French manufacturing. *R&D, patents, and productivity*, Zvi Griliches, (Eds.) University of Chicago Press, Chicago, IL. (1984): 375–392.

21. Xu J, Liu F, Chen Y. R&D, advertising and firms' financial performance in South Korea: does firm size matter? *Sustainability.* (2019) 11:3764. doi: 10.3390/su11143764

22. Lee IH, Marvel MR. The moderating effects of home region orientation on R&D investment and international SME performance: lessons from Korea. *Eur Manag J.* (2009) 27:316–26. doi: 10.1016/j.emj.2009.04.011

23. Yoo J, Lee S, Park S. The effect of firm life cycle on the relationship between R&D expenditures and future performance, earnings uncertainty, and sustainable growth. *Sustainability.* (2019) 11:2371. doi: 10.3390/su11082371

24. Koellinger P. The relationship between technology, innovation, and firm performance—empirical evidence from e-business in Europe. *Res Policy*. (2008) 37:1317–28. doi: 10.1016/j.respol.2008.04.024

25. What is advertising? Available at: https://economictimes.indiatimes.com/ definition/advertising (Accessed December 25, 2022).

26. Mirza HH, Hussain H, Gull W. Impact of advertising expenditure on firm performance: evidence from listed companies of Pakistan. *Int J Financ Eng.* (2022) 9:2250012. doi: 10.1142/S2424786322500128

27. Niu M, Ma W. Does corporate social responsibility heterogeneity affect corporate financial performance through technological innovation? The moderating effects of advertising intensity. *Front Psychol.* (2022) 13:837967. doi: 10.3389/fpsyg.2022.837967

28. Pourkarimi P, Kam E. The impact of R&D and advertising on firm performance in high-tech industries—evidence from the us information and communications technology industry. J Quant Econ. (2022) 20:723–53. doi: 10.1007/s40953-022-00302-1

29. Pal B, Nandy M. Innovation and business sustainability (IBS): empirical evidence from Indian pharmaceutical industry (IPI). *Artif Intell Eng Des Anal Manuf.* (2019) 33:117–28. doi: 10.1017/s0890060419000040

30. Woodside AG, Persing A, Ward B, DeCotiis AR. Advertising archetypes' impact on physician engagement and behavior in the context of healthcare products. *Psychol Mark.* (2018) 35:533–41. doi: 10.1002/mar.21104

31. Rahman M, Rodríguez-Serrano MÁ, Lambkin M. Advertising efficiency and profitability: evidence from the pharmaceutical industry. *Ind Mark Manage*. (2020) 89:619–29. doi: 10.1016/j.indmarman.2019.02.001

32. Schramm J, Andersen M, Vach K, Kragstrup J, Peter Kampmann J, Søndergaard J. Promotional methods used by representatives of drug companies: a prospective survey in general practice. *Scand J Prim Health Care.* (2007) 25:93–7. doi: 10.1080/02813430701339659

33. Elrod JK, Fortenberry JL. Advertising in health and medicine: using mass media to communicate with patients. *BMC Health Serv Res.* (2020) 20:818. doi: 10.1186/ s12913-020-05599-3

34. Guenzi P, Sajtos L, Troilo G. The dual mechanism of sales capabilities in influencing organizational performance. J Bus Res. (2016) 69:3707–13. doi: 10.1016/j.jbusres.2016.03.033

35. Nithya J, Com DM, Selvakumar N, Monisha SC. Intrinsic value of salesrersons and their value added attribute-a critical assessment. *J Pharm Negat Results*. (2022) 13:1209–19. doi: 10.47750/pnr.2022.13.S08.150

36. Chiplunkar S, Gowda D, Shivakumar H. Adaptation of pharmaceutical marketing and drug promotion practices in times of pandemic COVID-19. *Int J Health Allied Sci.* (2020) 9:11. doi: 10.4103/ijhas.IJHAS_81_20

37. Kakeesh DF, Al-Weshah GA, Dababneh RB. Leveraging market share through selling skills: the mediating role of adaptive selling behaviour. *Int J Bus Perform Manag.* (2021) 22:363–79. doi: 10.1504/ijbpm.2021.118599

38. Morgan C, Zane DM. Practitioner perspectives on key challenges in pharmaceutical marketing and future research opportunities. *J Public Policy Mark.* (2022) 41:368–82. doi: 10.1177/07439156221112304

39. Barbieri JC, Vasconcelos IFGD, Andreassi T, Vasconcelos FCD. Innovation and sustainability: new models and propositions. *Revista de Administração de Empresas*. (2010) 50:146–54. doi: 10.1590/S0034-75902010000200002

40. Kuzma E, Padilha LS, Sehnem S, Julkovski DJ, Roman DJ. The relationship between innovation and sustainability: a meta-analytic study. *J Clean Prod.* (2020) 259:120745. doi: 10.1016/j.jclepro.2020.120745

41. Jackson G, Apostolakou A. Corporate social responsibility in Western Europe: an institutional mirror or substitute? *J Bus Ethics.* (2010) 94:371–94. doi: 10.1007/s10551-009-0269-8

42. Bansal P, Song H. Similar but not the same: differentiating corporate sustainability from corporate responsibility. *Acad Manag Ann.* (2017) 11:105–49. doi: 10.5465/annals.2015.0095

43. Chelawat H, Trivedi IV. The business value of ESG performance: the Indian context. Asian J Bus Ethics. (2016) 5:195–210. doi: 10.1007/s13520-016-0064-4

44. Xie J, Nozawa W, Yagi M, Fujii H, Managi S. Do environmental, social, and governance activities improve corporate financial performance? *Bus Strategy Environ*. (2019) 28:286–300. doi: 10.1002/bse.2224

45. De Lucia C, Pazienza P, Bartlett M. Does good ESG lead to better financial performances by firms? Machine learning and logistic regression models of public enterprises in Europe. *Sustainability.* (2020) 12:5317. doi: 10.3390/su12135317

46. Zhang F, Qin X, Liu L. The interaction effect between ESG and green innovation and its impact on firm value from the perspective of information disclosure. *Sustainability.* (2020) 12:1866. doi: 10.3390/su12051866

47. Zhang D, Lucey BM. Sustainable behaviors and firm performance: the role of financial constraints' alleviation. *Econ Anal Policy*. (2022) 74:220–33. doi: 10.1016/j. eap.2022.02.003

48. Fatemi A, Glaum M, Kaiser S. ESG performance and firm value: the moderating role of disclosure. *Glob Financ J.* (2018) 38:45–64. doi: 10.1016/j.gfj.2017.03.001

49. Clarke T. Balancing the triple bottom line financial social and environmental performance (2001) 26:16–27. doi: 10.1177/030630700102600402,

50. Margaritis D, Psillaki M. Capital structure and firm efficiency. J Bus Finance Account. (2007) 34:1447–69. doi: 10.1111/j.1468-5957.2007.02056.x

51. Coad A, Segarrac A, Teruel M. Innovation and firm growth: does firm age play a role? *Res Policy*. (2016) 45:387–400. doi: 10.1016/j.respol.2015.10.015

52. Kijkasiwat P, Phuensane P. Innovation and firm performance: the moderating and mediating roles of firm size and small and medium enterprise finance. *J Risk Financ Manag.* (2020) 13:97. doi: 10.3390/jrfm13050097

53. Yao Y, Feng J, Yang Q. Impact of financing constraints on firm performance: moderating effect based on firm size. *Comput Intell Neurosci.* (2022) 2022:1954164. doi: 10.1155/2022/1954164

54. Ismail AI, Rose RC, Abdullah H, Uli J. The relationship between organizational competitive advantage and performance moderated by the age and size of firms. *Asian Acad Manag J.* (2010) 15:157–73.

55. Loderer C, Stulz R, Waelchli U. Firm rigidities and the decline in growth opportunities. *Manag Sci.* (2017) 63:3000–20. doi: 10.1287/mnsc.2016.2478

56. Kuncová M, Hedija V, Fiala R. Firm size as a determinant of firm performance: the case of swine raising. *AGRIS Papers Econ Inf.* (2016) 8:77–89. doi: 10.7160/ aol.2016.080308

57. Ibhagui OW, Olokoyo FO. Leverage and firm performance: new evidence on the role of firm size. North Am J Econ. (2018) 45:57–82. doi: 10.1016/j.najef.2018.02.002

58. Mubeen R, Han D, Abbas J, Raza S, Bodian W. Examining the relationship between product market competition and Chinese Firms performance: the mediating impact of capital structure and moderating influence of firm size. *Front Psychol.* (2022) 12:709678. doi: 10.3389/fpsyg.2021.709678

59. Dhawan R. Firm size and productivity differential: theory and evidence from a panel of us firms. *J Econ Behav Organ.* (2001) 44:269–93. doi: 10.1016/S0167-2681(00)00139-6

60. Fan X, Chen C, Yuan L. Centralization and firm performance: new evidence on the role of firm size. *Wirel Commun Mob Comput.* (2022) 2022:1–17. doi: 10.1155/2022/2233484

61. Fosu S. Capital structure, product market competition and firm performance: evidence from South Africa. *Q Rev Econ Finance*. (2013) 53:140–51. doi: 10.1016/j. qref.2013.02.004

62. Iqbal M, Javed F. The moderating role of corporate governance on the relationship between capital structure and financial performance. *Int J Res Business Soc Sci.* (2017) 6:89–105. doi: 10.20525/ijrbs.v6i1.624

63. Shahzad J, Ali P, Ahmad T, Ali S. Financial leverage and corporate performance: does financial crisis owe an explanation. *Pak J Stat Oper Res.* (2015) 11:65–88. doi: 10.1234/pjsor.v11i1.781

64. Mathur N, Tiwari SC, Sita Ramaiah T, Mathur H. Capital structure, competitive intensity and firm performance: an analysis of Indian pharmaceutical companies. *Manag Financ*. (2020) 47:1357–82. doi: 10.1108/MF-01-2020-0009

65. Competitive advantage strategies for healthcare industry. Available at: https:// www.mdpi.com/journal/ijerph/special_issues/Competitive_Healthcare (Accessed December 25, 2022)

66. Jin X, Zheng P, Zhong Z, Cao Y. The effect of venture capital on enterprise benefit according to the heterogeneity of human capital of entrepreneur. *Front Psychol.* (2020) 11:1558. doi: 10.3389/fpsyg.2020.01558

67. Wang S, Wang D. Exploring the relationship between ESG performance and green bond issuance. *Front Public Health.* (2022) 10:897577. doi: 10.3389/fpubh.2022.897577

68. Zhang D, Wang C, Dong Y. How does firm ESG performance impact financial constraints? An experimental exploration of the COVID-19 pandemic. *Eur J Dev Res.* (2022) 35:219–39. doi: 10.1057/s41287-021-00499-6

69. Miller D, Le Breton MI. Governance, social identity, and entrepreneurial orientation in closely held public companies. *Entrep Theory Pract.* (2011) 35:1051–76. doi: 10.1111/j.1540-6520.2011.00447.x

70. Minutolo MC, Kristjanpoller WD, Stakeley J. Exploring environmental, social, and governance disclosure effects on the S&P 500 financial performance. *Bus Strategy Environ*. (2019) 28:1083–95. doi: 10.1002/bse.2303

71. Walls JL, Berrone P, Phan PH. Corporate governance and environmental performance: is there really a link? *Strateg Manag J.* (2012) 33:885–913. doi: 10.1002/smj.1952

72. Yuanyuan C, Yahya F, Waqas M, Hongbo L. Do visionary-feedback seeking CEOS enhance firm sustainability through eco-innovation? A moderated mediation model. *Front Psychol.* (2022) 12:6350. doi: 10.3389/fpsyg.2021.750885

73. Zhao L, Rasoulinezhad E. How has the coronavirus outbreak affected economic poverty in different Asian regions? *Singapore Econ Rev.* (2021) 44:1–24. doi: 10.1142/S021759082144001X

74. Lee J, Yang H. Pandemic and employment: evidence from COVID-19 in South Korea. J Asian Econ. (2022) 78:101432. doi: 10.1016/j.asieco.2021.101432

75. Naqvi HA. Role of government policies to attain economic sustainability amid COVID-19 environment. *Front Environ Sci.* (2022) 10:983860. doi: 10.3389/ fenvs.2022.983860

76. Su L. The impact of coordinated development of ecological environment and technological innovation on green economy: evidence from China. *Int J Environ Res Public Health.* (2022) 19:6994. doi: 10.3390/ijerph19126994

77. Noh Y. Financial effects of open innovation in the manufacturing industry. *Manag Decis.* (2015) 53:1527–44. doi: 10.1108/MD-12-2014-0681

78. Bonaccorso SN, Sturchio JL. Direct to consumer advertising is medicalising normal human experience against. *BMJ*. (2002) 324:910–1. doi: 10.1136/ bmj.324.7342.910

79. Fortenberry JL Jr, Elrod JK, McGoldrick PJ. Is billboard advertising beneficial for healthcare organizations? An investigation of efficacy and acceptability to patients. *J Health Manag.* (2010) 55:81–96. doi: 10.1097/00115514-201003000-00004

80. Jong G, BHC S, Sturkenboom MC. Marketing in the lay media and prescriptions of terbinafine in primary care: dutch cohort study. *BMJ*. (2004) 328:931. doi: 10.1136/bmj.38007.711481.F7

81. Elrod JK, Fortenberry JL. Formulating productive marketing communications strategy: a major health system's experience. *BMC Health Serv Res.* (2018) 18:926. doi: 10.1186/s12913-018-3676-7

82. Mintzes B. Direct to consumer advertising is medicalising normal human experience. *BMJ.* (2002) 324:908–11. doi: 10.1136/bmj.324.7342.908

83. Holden A, Nanayakkara S, Skinner J, Spallek H, Sohn W. What do Australian health consumers believe about commercial advertisements and testimonials? A survey on health service advertising. *BMC Public Health*. (2021) 21:74. doi: 10.1186/s12889-020-10078-9

84. Janodia MD, Udupa N. Marketing and advertising of prescription and over the counter (OTC) products – ethical issues. *Indian J Pharm Educ Res.* (2007) 18:383–7.

85. Francer J, Izquierdo JZ, Music T, Narsai K, Nikidis C, Simmonds H, et al. Ethical pharmaceutical promotion and communications worldwide: codes and regulations. *Philos Ethics Humanit Med.* (2014) 9:1–17. doi: 10.1186/1747-5341-9-7

86. Köhler C, Mantrala MK, Albers S, Kanuri VK. A meta-analysis of marketing communication carryover effects. *J Mark Res.* (2017) 54:990–1008. doi: 10.1509/jmr.13.0580

87. Adeoye S, Bozic KJ. Direct to consumer advertising in healthcare: history, benefits, and concerns. *Clin Orthopaed Rel Res.* (2007) 457:96–104. doi: 10.1097/BLO.0b013e31803427e6

88. Khan MS, Bory S, Rego S, Suy S, Durrance-Bagale A, Sultana Z, et al. Is enhancing the professionalism of healthcare providers critical to tackling antimicrobial resistance in low-and middle-income countries? *Hum Resour Health*. (2020) 18:10. doi: 10.1186/s12960-020-0452-7

89. Grundy Q, Hutchison K, Johnson J, Blakely B, Clay-Wlliams R, Richards B, et al. Device representatives in hospitals: are commercial imperatives driving clinical decision-making? *J Med Ethics*. (2018) 44:589–92. doi: 10.1136/medethics-2018-104804

90. Shaalan AS, Reast J, Johnson D, Tourky ME. East meets west: toward a theoretical model linking Guanxi and relationship marketing. *J Bus Res.* (2013) 66:2515–21. doi: 10.1016/j.jbusres.2013.05.043

91. Du M, Gao H, Zhang J. Toward a Guanxi-bases view of structural holes in sales gatekeeping: a qualitative study of sales practices in China. *Ind Mark Manage*. (2019) 76:109–22. doi: 10.1016/j.indmarman.2018.08.001

92. Ma B, Zhang J. How to leverage relational gatekeepers to improve sales performance: moderating effects of gatekeeper-salesperson Guanxi and gatekeeperbuyer Guanxi. *Ind Mark Manage*. (2022) 107:108–19. doi: 10.1016/j. indmarman.2022.09.018

93. Sawad AB, Andrews K. Marketing training strategies that pharmaceutical sales managers use to reduce unethical behavior. *J Educ Health Promot*. (2022) 11:202. doi: 10.4103/jehp.jehp_1081_21

94. Yi H, Amenuvor FE. The effect of door-to-door salespeople's individual sales capabilities on selling behavior and performance: the moderating effect of competitive intensity. *Sustainability*. (2022) 14:3327. doi: 10.3390/su14063327

95. Bhattacharya A, Good V, Sardashti H. Doing good when times are bad: the impact of CSR on brands during recessions. *Eur J Mark*. (2020) 54:2049–77. doi: 10.1108/EJM-01-2019-0088

96. Lee MT, Raschke RL, Krishen AS. Signaling green! Firm ESG signals in an interconnected environment that promote brand valuation. *J Bus Res.* (2022) 138:1–11. doi: 10.1016/j.jbusres.2021.08.061

97. Khatib FAS, Nour IA. The impact of corporate governance on firm performance during the COVID-19 pandemic: evidence from Malaysia. *J Asian Financ Econ Business*. (2021) 8:943–52. doi: 10.13106/jafeb.2021.vol8.no2.0943

98. Asa'D Abu IA, Nour A, Atout S. The impact of financial performance on firm's value during COVID-19 pandemic for companies listed in the Palestine exchange (2019–2020). A.M.A. Musleh Al-Sartawi, A. Razzaque and M.M. Kamal (Eds.) *From the internet of things to the internet of ideas: The role of artificial intelligence*, Cham: Springer (2022). 529–551.